Author's response to reviews

Title: Retention of foreign body in the gastrointestinal tract: a sign of congenital obstructive anomaly

Authors:

Pravas C Subudhi (beam.diagnostics@gmail.com)
Shivaram P Singh (sudhasingh@sify.com)
Chudamani Meher (beam.diagnostics@gmail.com)
Omprakash Agrawal (beam.diagnostics@gmail.com)

Version: 2 Date: 17 December 2007

Author's response to reviews:

We believe the paper may be of particular interest to the readers of your journal as it provides a new potential way to treat nonalcoholic Steatohepatitis.

The above sentence in covering letter should be changed to:

We believe the paper may be of particular interest to the readers of your journal as it highlights the fact that retention or non passage of a radio-opaque foreign body in a child should warn the pediatrician to the possibility of an obstructive congenital anomaly in the child.